Author's response to reviews

Title: Detection and treatment of an aneurysma spurium of the arteria hepatica dextra after laparoscopic cholecystectomy

Authors:

Oliver Kirschberg (oliver.kirschberg@web.de)
Andreas Scheding (andreas.scheding@vivantes.de)
Thomas Saers (saerst@kliniken-koeln.de)
Bernd Krakamp (krakampb@kliniken-koeln.de)

Version: 3 Date: 8 July 2013

Author's response to reviews: see over
Cover letter

MS: 7471527829070021 – Detection and treatment of an aneurysma spurium of the arteria hepatica dextra after laparoscopic cholecystectomy

Editorial Requirements

As the authors, we would like to thank the editors very much for giving us the constructive feedback on our manuscript. We certainly would like to improve the quality of the language and increase the clarity of the case presentation. We asked a colleague from California, U.S., to read and edit the manuscript carefully and to improve the English in the manuscript. The revised manuscript conforms to the journal style.

All authors have read and agreed to the final version of the revised manuscript.

With the best wishes

Oliver Kirschberg